Author’s response to reviews

Title: The FATZO mouse, a next generation model of type 2 diabetes, develops NAFLD and NASH when fed a western diet supplemented with fructose

Authors:

Gao Sun (gaogao_36@hotmail.com; gao.sun@crownbio.com)
Charles Jackson (vjackson@crownbio.com)
Karen Zimmerman (kzimmerman@crownbio.com)
Li-Kun Zhang (zhanglikun@crownbio.com)
Courtney Finnearty (cmfinnea@umail.iu.edu)
George Sandusky (gesandusky@umail.iu.edu)
Guo-Dong Zhang (guodong.zhang@crownbio.com)
Richard Peterson (rpeterson@crownbio.com)
Yi-Xin Wang (yxwang@crownbio.com)

Version: 1 Date: 10 Jul 2018

Author’s response to reviews:

Dear Madam/Sir

I have received the request for the revision of the manuscript to match the formatting requirement of the journal before further reviewing process can be proceeded.

I have now re-attached the revised manuscript and ARRIVE checklist as a supplementary document for your reference.

Please let me know if there is any other problems with the formatting of the manuscript.

Best regards

Gao Sun